Author's response to reviews

Title: Mycobacterium tuberculosis Complex Drug Resistance Pattern and Identification of Species Causing Tuberculosis in the West and Centre Regions of Cameroon

Authors:

Jean-Paul Assam-Assam (assamjean@yahoo.fr)
Veronique B Penlap (v.penlap@yahoo.fr)
Fidelis Cho-Ngwa (chongwa_ub@yahoo.co.uk)
Jean-Claude Tedom (tedomjc@yahoo.fr)
Irene Ane-Anyangwe (ianyangwe@yahoo.com)
Vincent P Titanji (vpk.titanji@yahoo.com)

Version: 4 Date: 11 April 2011

Author's response to reviews:

Dear Sir/Madam,

Please consider the following revision:

Title: We changed ...in two regions of Cameroon to ...in the West and Centre regions of Cameroon

Author list: we wrote the names of authors exactly as they are in the submission system, in style and order

Text: we formatted the references within text to use square brackets instead of numbers in superscripts; Also, we changed all coloured text to black.

Tables: we removed the visible vertical lines from the tables

Kind regards,
Fidelis